Author’s response to reviews

Title: A randomised controlled trial to compare opt-in and opt-out parental consent for childhood vaccine safety surveillance using data linkage: study protocol

Authors:

Jesia G Berry (Jesia.Berry@adelaide.edu.au)
Philip Ryan (Philip.Ryan@adelaide.edu.au)
Annette J Braunack-Mayer (Annette.BraunackMayer@adelaide.edu.au)
Katherine M Duszynski (Katherine.Duszynski@adelaide.edu.au)
Vicki Xafis (Vicki.Xafis@adelaide.edu.au)
Michael S Gold (Michael.Gold@adelaide.edu.au)
Vaccine Assessment Using Linked Data (VALiD) Working Group (valid.study@adelaide.edu.au)

Version: 6 Date: 20 December 2010

Author’s response to reviews: see over
20 December 2010
Trials

Dear Editorial Production Team,

RE: MS: 5821322344488407. Changes required: Due 21 December 2010

I have identified a few minor typographical changes to be made, which are highlighted using track changes in the uploaded revised manuscript. The manuscript is otherwise acceptable for publication and we look forward to seeing it published.

Yours sincerely
Jesia Berry,
Corresponding Author
Email: jesia.berry@adelaide.edu.au